Author's response to reviews

Title: Violence, mental health and violence risk factors among community women: an epidemiological study based on two national household surveys in the UK

Authors:

Min Yang Dr (min.yang@nottingham.ac.uk)
Stephen C P Wong Dr (s.wong@sasktel.net)
Jeremy W Coid Dr (j.w.coid@qmul.ac.uk)

Version: 2 Date: 19 December 2012

Author's response to reviews:

In this version we have included a title page with information formatted according to the submission guidelines. We have also presented details of the approval of the ethics committee for the survey data used in the present study in the Method section. Finally an proof reading of English was undertaken by a fluent English speaking colleague.